Author's response to reviews

Title: Unexpected decline in tuberculosis cases coincident with economic recession -- United States, 2009

Authors:

Carla A Winston (cwinston@cdc.gov)
Thomas R Navin (tnavin@cdc.gov)
Jose E Becerra (ibecerra@cdc.gov)
Michael P Chen (mchen1@cdc.gov)
Lori R Armstrong (lararmstrong@cdc.gov)
Carla Jeffries (cjeffries@cdc.gov)
Rachel S Yelk Woodruff (rwoodruff@cdc.gov)
Jessie Wing (jwing@cdc.gov)
Angela M Starks (astarks@cdc.gov)
Craig M Hales (chales@cdc.gov)
J Steve Kammerer (skammerer@cdc.gov)
William R Mac Kenzie (wmackenzie@cdc.gov)
Kiren Mitruka (kmitruka@cdc.gov)
Mark C Miner (mminer@cdc.gov)
Sandy Price (sprice@cdc.gov)
Joseph Scavotto (jscavotto@cdc.gov)
Ann M Cronin (acronin@cdc.gov)
Phillip Griffin (pgriffin@kdheks.gov)
Philip A LoBue (plobue@cdc.gov)
Kenneth G Castro (kcastro@cdc.gov)

Version: 5 Date: 14 October 2011

Author's response to reviews:

*October 14 2011 version responses to October 4 2011 BMC proofreading comments.

Formatting changes to be made:

-----------------------------------

Major revisions (we require the author to make these changes)

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.
The preferred style is "First name Initial Last name" (e.g. Joe F Bloggs)
*synchronized author middle initials in text and submission system

Authors’ contributions: Please indicate that all authors read and approved the final manuscript.
*confirmed and indicated under contributions

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript, and change any coloured text to black

*completed

Please remove "Key words"

*done

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.